**RESEARCH ARTICLE** 

## Neutral Models of Microbiome Evolution

#### Qinglong Zeng, Jeet Sukumaran<sup>\*a</sup>, Steven Wu<sup>\*b</sup>, Allen Rodrigo\*

Biology Department, Duke University, Durham, North Carolina, United States of America

¤a Current address: Department of Ecology and Evolutionary Biology, University of Michigan, Ann Arbor, Michigan, United States of America

¤b Current address: The Biodesign Institute, Arizona State University, Tempe, Arizona, United States of America

\* a.rodrigo@duke.edu

## Abstract

There has been an explosion of research on host-associated microbial communities (i.e., microbiomes). Much of this research has focused on surveys of microbial diversities across a variety of host species, including humans, with a view to understanding how these microbiomes are distributed across space and time, and how they correlate with host health, disease, phenotype, physiology and ecology. Fewer studies have focused on how these microbiomes may have evolved. In this paper, we develop an agent-based framework to study the dynamics of microbiome evolution. Our framework incorporates neutral models of how hosts acquire their microbiomes, and how the environmental microbial community that is available to the hosts is assembled. Most importantly, our framework also incorporates a Wright-Fisher genealogical model of hosts, so that the dynamics of microbiome evolution is studied on an evolutionary timescale. Our results indicate that the extent of parental contribution to microbial availability from one generation to the next significantly impacts the diversity of microbiomes: the greater the parental contribution, the less diverse the microbiomes. In contrast, even when there is only a very small contribution from a constant environmental pool, microbial communities can remain highly diverse. Finally, we show that our models may be used to construct hypotheses about the types of processes that operate to assemble microbiomes over evolutionary time.

#### Author Summary

Microbial communities associated with animals and plants (i.e., microbiomes) are implicated in the day-to-day functioning of their hosts. However, we do not yet know how these host-microbiome associations evolve. In this paper, we develop a computational framework for modelling the evolution of microbiomes. The models we use are neutral, and assume that microbes have no effect on the reproductive success of the hosts. Therefore, the patterns of microbiome diversity that we obtain in our simulations require a minimal set of assumptions relating to how microbes are acquired and how they are assembled in the environment. Despite the simplicity of our models, they help us understand the patterns seen in empirical data, and they allow us to build more complex hypotheses of hostmicrobe dynamics.



## 

**Citation:** Zeng Q, Sukumaran J, Wu S, Rodrigo A (2015) Neutral Models of Microbiome Evolution. PLoS Comput Biol 11(7): e1004365. doi:10.1371/ journal.pcbi.1004365

Editor: Claus O. Wilke, University of Texas at Austin, UNITED STATES

Received: January 21, 2015

Accepted: May 18, 2015

Published: July 22, 2015

**Copyright:** © 2015 Zeng et al. This is an open access article distributed under the terms of the <u>Creative Commons Attribution License</u>, which permits unrestricted use, distribution, and reproduction in any medium, provided the original author and source are credited.

Data Availability Statement: Empirical data from the human microbiome were obtained from the website of the NIH Human Microbiome Project (http://www. hmpdacc.org/HMSMCP/). As is described on the HMP website, all the samples were collected from 16 body sites from 103 healthy humans and processed with Whole Genome Shotgun sequencing. Specific information of each sample is available on the website (http://www.hmpdacc.org/HMIWGS/all/). We selected data of microbial communities associated with anterior nares, buccal mucosa, supragingival plaque, stool, tongue dorsum and vaginal posterior formix because of their large numbers of samples, and removed the replicated samples from the same human subject for each body site (see <u>S1 Dataset</u>). The HMP data provides relative abundances for different taxonomic levels. We chose to use genuslevel diversities. The relative abundances of genera for each site and all samples is given in <u>S1 Dataset</u>. All simulations were carried out using Python scripts and Java programs, available from <u>https://github.com/ gz28/microbiosima.git</u>.

Funding: AR was partially supported by the US National Science Foundation through a grant awarded to the National Evolutionary Synthesis Center. SW was partially supported by NIH Grants R01-GM101352 to RA Zufall, RBR Azevedo, and RA Cartwright and R01-HG007178 to DF Conrad and RA Cartwright. The funders had no role in study design, data collection and analysis, decision to publish or preparation of the manuscript.

**Competing Interests:** The authors have declared that no competing interests exist.

#### Introduction

Microbial communities associated with animals and plants (i.e., microbiomes) are implicated in the day-to-day functioning of their hosts in a variety of ways; microbes provide hosts with access to nutrients [1–4], they protect against pathogens [5,6], confer drought resistance [7], mediate hosts' social interactions [8], and modulate host behavior [9–11]. Not surprisingly, many studies have related human health to the human microbiome. Researchers have proposed that perturbations in human microbiome composition may be associated with a range of disorders, including inflammatory bowel disease [12–14], anxiety and depression [15], obesity [16,17], autism [18,19], allergic responses [20] and respiratory ailments [21–23]. There is a growing number of large-scale projects underway to characterize and analyze the collection of microbes linked to human health and disease using advanced sequencing methods and technologies, including the Human Microbiome Project [24] and the European Metagenomics of the Human Intestinal Tract [25].

Many microbiome studies focus on descriptions of microbial dynamics within hosts over short periods [26-28], or on the composition and diversity of microbial communities amongst hosts with different phenotypes or under different treatment regimes [17,29], but there is increasing recognition that host-microbe dynamics need to be studied within the context of a unified ecological or evolutionary framework. In a recent paper, Costello et al [30] listed four ecological processes that mediate the diversity of human microbiota: environmental selection, whereby the host environment favors the presence and persistence of certain microbial taxa; historical contingency, in which differences in the timing and order of microbial colonization leads to differences in succession and climax communities; random sampling, where stochastic factors influence community assemblages; and dispersal limitation, where the availability of microbial taxa is restricted by the local structure of host communities and environments. Yeoman et al [31] discuss many of these same factors, and frame the challenge of understanding microbial diversity through the lens of evolution, identifying selection induced by competition between individuals of the same or different microbial species, and host influences including phylogeny, as important drivers of host-microbial variation.

In this paper, we take the first steps in developing an explicit framework for modelling the evolutionary and ecological dynamics of microbial communities within a population of hosts. The models we use are related to some of those that metacommunity theorists work with [32,33], specifically, neutral models of ecology and biodiversity [34], but there is one significant difference. Standard spatial models of community assembly, biogeography and biodiversity assume that areas available for colonization remain indefinitely and are static. These models do not apply to the evolution of microbiomes because the "spaces" these microbial communities colonize-that is, the hosts-share an evolutionary history characterized by lineages that persist or go extinct. For this reason, the pairing of host genealogy with microbiome assembly may lead to different patterns of biodiversity than those expected under existing models of microbial community assembly and metacommunity theory.

We develop an agent-based framework, similar to that used recently by Hellweger et al [35], to capture the emergent patterns of microbial diversity over many host generations. In contrast to Hellweger et al, we do not model mutations and speciation in microbial lineages but focus instead on how ecological and evolutionary sampling processes affect the standing variation of microbes in the environment and hosts. The neutral processes that are encapsulated in our framework are based on those that have been proposed by others [30,31] with the added dimensions of an evolutionary timescale and a genealogy of hosts, both of which contribute to shaping the eventual composition and variation of microbial communities within and between

hosts. Our models do not assume that hosts differ in their reproductive success as a consequence of their microbial communities; nor do our models assume differences amongst microbes in their propensities to persist in the host or the environment, or in their abilities to disperse. Consequently, whereas our models do not capture the complexity of processes that are likely to mediate the evolution of microbiomes, they serve as minimalist null models against which empirical patterns may be compared.

Our framework is a generic one, and we have not developed it specifically for any one species of host, or indeed, any one systemic compartment within a host. Instead, our models incorporate simple host mechanisms for microbial acquisition, and we explore the effects of parental inheritance of the microbiome, and microbial recruitment from the environment. Our results indicate that parental inheritance tends to reduce microbial diversity and increase homogeneity within hosts, while ongoing environmental acquisition works to maintain microbial heterogeneity within hosts. Interestingly, we observe a non-linear relationship between the degree of parental inheritance and between-host differences in microbiome composition. Finally, we show how these neutral patterns allow us to make predictions about the processes that are important in shaping microbiome diversity when applied to empirical data.

#### The models

Our framework applies to a population of hosts and an available pool of microbial colonists. As a first step, we assume that hosts do not exert any preferences on the microbial taxa they acquire. Similarly, we assume that microbes do not interfere with host reproductive capacity, or the survivorship and reproductive success of other microbes in the community. As will become clear, the only indirect effects that influence microbial recruitment and persistence from one generation of hosts to the next are competition for space within hosts and the relative abundance of microbial taxa. Simply put, we assume that the ecological and evolutionary processes that operate on hosts and their microbiomes are neutral; in this regard, our framework is analogous to neutral theories in evolutionary biology [36,37], ecology and biodiversity [34]. We expand on this analogy later, but for now, we note that neutral theories provide parsimonious accounts of the types of patterns that can emerge in complex systems, they serve as null models for statistical hypothesis tests, and they provide platforms upon which we may construct more elaborate representations of these same systems [38].

In this framework, hosts reproduce as exually in discrete generations, following a neutral Wright-Fisher process [39,40], where each individual in a succeeding generation chooses a parent randomly from the preceding generation. Hence, with a population of hosts of constant size *N*, all as exual individuals will share a common ancestor after 2*N* generations, on average. In our models, as exual reproduction is a computational convenience, and can be replaced with sexual reproduction without changing the essential patterns that we observe.

We model how hosts acquire their microbiomes in three ways (Fig.1). First, under a strict "parental-acquisition" (PA) process, all hosts acquire their microbial communities directly from their parents. Second, with strict "environmental-acquisition" (EA), hosts acquire their microbiomes solely from the environment. Between these two extremes, we also allow a third "mixed-acquisition" (MA<sub>x</sub>) process, whereby hosts acquire some percentage, x%, of their microbiomes from their parents and (100-x)% from the environment. MA<sub>0</sub> is exactly equivalent to EA, and MA<sub>100</sub> to PA; as such, EA and PA designate boundary conditions of the ecological processes that mediate microbial acquisition in hosts.

It is worth pausing at this point to clarify what we mean when we say that hosts acquire their microbiomes from their "parents" or their "environments". Our models do not explicitly take account of the life events–illness, infections, changes in environments or diets–of each



**Fig 1. The relationship among parents, offspring and environment in different environmental and acquisition models.** The columns represents different models of the environmental pool; the rows represents different models of microbiome acquisition. Arrows indicate how the environmental community is assembled for each new generation, and how offspring acquire their microbiomes. The values of *y*% and *x*% represent the percentage of host contribution under mixed environment (ME), and the percentage of parental inheritance under mixed acquisition (PA), respectively.

doi:10.1371/journal.pcbi.1004365.g001

host within a generation, nor does it consider microbial fluxes within the lifespan of each host. Instead, the microbial composition of each host is essentially measured as an aggregate over the single generation that the host exists. Consequently, when we quantify the percentage of microbes from parents and environment using, say,  $MA_{10}$ , we mean that over the life of the host, 90% of its microbes come from the environment and 10% from its parent. In our models, it is possible that the parental contribution happened in the first 10% of the host's life, or it may be that over the entire lifespan of the host, there was an ongoing contribution by the parent that amounted to 10% of the microbial composition.

Since we allow hosts to recruit microbes from an "environment", we need to define how the microbial content of this environment is constituted. In simulations, we characterize microbial composition using a distribution of taxa' relative abundances. We propose three processes that determine the composition of the pool of microbes available for recruitment. First, we assume that the environment has a microbial composition that remains fixed over time. For the "fixed environment" (FE), all taxa are present in the environment throughout the simulation, and are available to every generation of hosts. The second process we propose involves a changing environmental microbial profile, whereby the relative abundance of each microbial taxon available to the hosts in a given generation, is an aggregate of their abundances from all hosts of the

preceding generation. Under this "pooled-environment" (PE), microbial composition is reflection of what was present in the parents of the current generation of hosts. A third, intermediate, process is a combination of the previous two "environments": the environmental microbial pool available for recruitment contains a percentage, *y*%, from the parental pool of microbes, and (100-y)% of microbes from the fixed environment. Under this "mixed environment" (ME<sub>*y*</sub>), the proportion of contribution from host microbiomes is given by *y*. As with our acquisition models, ME<sub>0</sub> and ME<sub>100</sub> are equivalent to the boundaries FE and PE, respectively.

Our framework allows us to combine different host-acquisition processes with different ways of constructing the pool of available microbes in the environment. Conceptually, each of these combinations is a particular neutral model, capturing some of the elements previously discussed in the literature. For instance, PA or  $MA_x$  incorporate the phylogenetic dependencies that Yeoman et al [31] discuss, and EA x PE is equivalent to what Costello et al [30] call dispersal limitation, whereby the local host community influences microbial composition. It is worth noting that the combinations PA x (FE,  $ME_y$ , PE)–read as "PA in combination with FE, with  $ME_y$  or with PE"–will give identical results. This is because, in all cases, the environment contributes nothing to host microbial content (see the first row in Fig 1).

We model the construction of the microbial community in each host by competitive random sampling with replacement. Under this process, each host allows only a fixed and limited number of microbes to populate its microbiome. If microbial acquisition occurs under EA, each host samples randomly from the available pool of taxa according to the relative abundance of each taxon in the environment. In the case of  $MA_x$ , x% of microbes are selected from the parent and (100-x)% from the environment. If hosts acquire their microbial taxa under PA, then all microbes are inherited from the hosts' parents, although the relative abundance of each taxon fluctuates multinomially. By constructing microbial communities in this way, we allow stochastic factors and indirect competition to modify taxon composition within and between hosts, as proposed by Costello et al [30].

By simulating combinations of PA,  $MA_x$  and EA against FE,  $ME_y$  and PE forward in time over many host generations and over a range of conditions, we are able to recover data on the behavior of individual microbial taxa, as well as a variety of summary statistics, including the expected time it takes individual taxa to invade all hosts or go extinct in the host population, and the trajectories of microbial taxonomic richness (measured simply as the total number of microbial taxa) and microbial taxonomic evenness (measuring the similarity in the frequency of each taxon), microbial diversity within hosts ( $\alpha$ -diversity), inter-host variation in microbial composition ( $\beta$ -diversity) and the aggregate microbial diversity from all hosts in the population ( $\gamma$ -diversity). Here, we report only on the latter three measures of diversity.

#### Results

#### Relative abundances of microbial taxa in the host population

Microbial diversity within the host population is a function of the proportion of microbes that parents contribute directly to offspring and the proportion they contribute to the environment. Fig 2 illustrates how population-level taxon abundances change under various combinations of these proportions. In our simulations, the distributions of taxon abundances under high levels of parental contributions are skewed, and may be approximated by commonly-applied distributions, including the log-normal distribution and the Dirichlet multinomial (DM) distribution [41] (Fig 3; the DM distribution has the advantage of allowing  $\alpha$ -,  $\beta$ - and  $\gamma$ -diversities to be simulated–see S2 Fig). The ability to recover skewed abundance distributions is interesting, because we begin our simulations with a uniform distribution of microbial taxa, and we retain this uniform distribution in the fixed environment throughout the evolutionary history of the





**Fig 2. Distribution of relative taxon abundance within the host population under combinations of MA and ME.** (a) distribution of relative taxon abundances with different parental contributions to the environment (ME), when the proportion of direct parental acquisition of microbes is low (i.e., MA(50) or 50% parental acquisition); (b) distribution of relative taxon abundances with different parental acquisition); (b) distribution of relative taxon abundances with different parental acquisition of microbes is high (i.e., MA(98.4) or 98.4% parental acquisition); (c) distribution of relative taxon abundances with different parental acquisition of relative taxon abundances with different proportions of microbes directly acquired from parents (PA), when the proportion of parental microbial contribution to the environmental pool is low (i.e., ME (50) or 50% parental contribution); (d) distribution of relative taxon abundances with different proportions of microbes directly acquired from parents (PA), when the proportion of parental microbial contribution); (d) distribution to the environmental pool is high (i.e., ME(99.9) or 99.9% parental contribution).

doi:10.1371/journal.pcbi.1004365.g002

host population. Consequently, the emergence of dominant and rare taxa is a consequence of repeated parental contributions either directly to the next generation of hosts or indirectly to the environment.



**Fig 3. Log-normal and Dirichlet-Multinomial fitting of simulated abundance distributions.** Relative frequency histograms represent the abundance distributions of simulated microbial communities under two mixed models. The red lines are the probability density curves of fitted log-mormal distributions. The green lines are smoothed density curves from Dirichlet-Multinomial distributions.

doi:10.1371/journal.pcbi.1004365.g003

In fact, all simulations in which there was complete parental acquisition of microbes (i.e., PA) resulted in the loss of all but one microbial taxon in the host population. Similarly, when the environment was reconstituted each generation exclusively with microbes from the parents (i.e., PE), the same pattern was observed with only a single microbial taxon remaining. These result are consistent with predictions made under neutral models of community ecology [42], and highlight the strong depressive effect of parental transmission, either directly from parent to offspring or via parental contributions to a local pool of microbes, on population-level microbiome diversity.

With EA x FE, microbes are obtained randomly from a fixed environment that persists over the evolutionary history of the hosts; unsurprisingly, the host population retains all microbes found in the environment. Interestingly, when microbes are obtained both from parents and a fixed environment (MA x FE), we still see the persistence of all or almost all microbes in the host population (see <u>Fig 2A and 2B</u>, first column of each bar chart; ME(0) is equivalent to a fixed environment with no microbial contributions from parents). This is true even when the proportion of microbial taxa that an individual host acquires from the fixed environment at each generation is very small, on the order of 0.001. Therefore, a very small contribution from a constant environmental source of microbes is sufficient to retain high levels of microbial diversity in the host population.

#### Measures of diversity

Microbial diversities are frequently measured in three ways:  $\alpha$ -diversity,  $\beta$ -diversity, and  $\gamma$ diversity. Our simulations indicate that all three measures depend on the percentage of parental contribution to offspring microbiomes and the composition of the environmental microbial pool (see <u>S1-S3</u> Tables for simulation means and standard deviations).

Under our neutral model, in which the absence of host sub-population structure means that all hosts sample their microbes from the same environment,  $\alpha$ - and  $\gamma$ -diversities remain high, and  $\beta$ -diversity remains low, for a large part of the range of direct or indirect parental





# **Fig 4. Heatmap of α-, γ-, β-diversity with different combinations of acquisition models and environmental models.** For each heatmap, the proportion of parental inheritance increases and environmental acquisition decreases from left to right; similarly, the environment microbial composition is increasingly derived from the parental microbiomes as we move from the bottom of each heatmap to the top. Each diversity value is averaged from 10 independent simulations.

doi:10.1371/journal.pcbi.1004365.g004

contributions (i.e., to offspring or to the environment, respectively). Nonetheless, at high values of parental contributions, there are discernible differences in diversities, and we have also focused our simulations in these areas (Fig 4; see S4–S6 Tables for simulation means and standard deviations). In general,  $\alpha$ -diversity (average diversity within hosts) and  $\gamma$ -diversity (overall diversity within the entire population of hosts) increase as we increase the fixed environmental contribution because a fixed environment helps maintain a uniform distribution of taxon abundances and delays the loss of microbial taxa during evolution. Conversely, when hosts acquire increasing proportions of their microbiomes from their parents directly, or indirectly from a pooled environment, the variation of taxon abundance increases and taxon richness tends to decrease, thus lowering both  $\alpha$ - and  $\gamma$ -diversities (Fig 4A and 4B; S1, S2, S4 and S5 Tables). Inter-host variation in microbial composition, or  $\beta$ -diversity, also depends on the degree of parental inheritance, and the ratio of fixed-to-pooled environmental components (Fig 4C; S3 and S6 Tables).

Under the combination of PA x (FE, ME or PE),  $\beta$ -diversity tends to zero, because all hosts descend from a single common ancestor and, as noted above, only a single microbial taxon remains in all hosts. When we have the parental microbiome as the only source of microbiomes in the next generation, ultimately, all lineages will have acquired their microbiomes from the most recent common ancestor (MRCA) of the population of hosts. Additionally, from one generation to the next, stochastic sampling of microbes over evolutionary time will result in the loss of all but one microbial taxon. Interestingly, with a high percentage of environmental acquisition,  $\beta$ -diversity is also relatively low, because all hosts acquire a large proportion of their microbial taxa from the same environmental pool, and consequently, will tend to acquire the same set of taxa. As noted above, the highest  $\beta$ -diversity occurs in a relatively narrow range of values of pure parental acquisition (between 87–99% of direct parental transmission; <u>S6 Table</u>).

If we focus on the relationship between  $\beta$ -diversity and the environmental pool, we see that its behavior is similar to that of  $\alpha$ - and  $\gamma$ -diversities: it decreases as we increase the pooled

environmental contribution to offspring microbiome. This is because a pooled environment, with contributions from the parental generation, tends to give rise to a non-uniform distribution of microbial taxa. As the degree of parental contribution increases, the environmental community will be dominated by few highly abundant species which are likely shared by most or all hosts within the population, accounting for high between-host similiarity in microbial composition (Fig 4C; S3 and S6 Tables).

It is important to note that our simulations have not been performed with inference or prediction in mind: the number of hosts, the number of microbes, and the number of taxa in our simulations are not necessarily equivalent to those of real-world microbial communities and their hosts, nor have we necessarily chosen the appropriate diversity indices or taxonomic resolution to optimize prediction/inference. Nonetheless, it is helpful to examine how the simulated values of diversity compare to empirical observations, and what these comparisons might tell us about the evolutionary processes that are acting on microbiomes. As an example, we used genus-level taxonomic data from the NIH Human Microbiome Project (HMP) [43], specifically, a table of relative abundance found in different compartments of the human body (http://www.hmpdacc.org/HMSMCP/; see S1 Dataset). Several large samples from the anterior nares, vaginal posterior fornix, stool, buccal mucosa, tongue dorsum and supragingival plaque were chosen to calculate  $\alpha$ -,  $\beta$ - and  $\gamma$ -diversities on genus level (Table 1).

Values of  $\alpha$ - and  $\gamma$ -diversity obtained from all sampled sites of the human microbiome are low, in comparison to most of the values we obtained in our simulations. In fact, human microbiome diversities are generally lower than those of other non-human primates [44,45]. If we compare the empirical diversities to those obtained in our simulations (S1–S6 Tables), we would have to posit very high parental contributions, both direct and indirect (>90%), to account for the  $\alpha$ - and  $\gamma$ -diversities across all human body sites. In contrast, values of  $\beta$ -diversity appear to provide a little more discrimination amongst body sites: the site with the lowest  $\beta$ -diversity is the vaginal posterior fornix, and its value is consistent with a very low degree of direct parental contribution in our simulations (approximately between 0–15%). The  $\beta$ -diversities at other sites appear to suggest higher levels of parental contribution (again, >90%). In the next section, we discuss the implications of these results, as they relate to human microbiome evolution and how the neutral model may be used to construct hypotheses about relevant evolutionary and ecological processes.

#### Discussion

In this paper, we introduce a simple and flexible framework to model the evolution of microbiomes within a population of hosts, which takes account of different modes of microbiome

Body Site	α-Diversity <sup>1</sup>	γ-Diversity <sup>2</sup>	β-Diversity	Number of samples
Posterior Fornix	0.041±0.067	0.097	0.175±0.294	38
Stool	0.267±0.093	0.381	0.464±0.213	94
Supragingival Plaque	0.457±0.061	0.531	0.455 ±0.136	85
Tongue Dorsum	0.399±0.051	0.454	0.387±0.137	87
Anterior Nares	0.223 ±0.071	0.318	0.451 ±0.210	69
Buccal Mucosa	0.265±0.079	0.315	0.318 ±0.135	77

Table 1. Diversities of microbiota associated with different human body sites.

<sup>1</sup>The  $\alpha$ - and  $\beta$ - diversities are mean values and standard errors averaged over all samples or over all pairs of samples.

<sup>2</sup>The γ- diversity is a single value because it is calculated with a large sample that is pooled from all the samples within the same body site; hence no replicates or ranges are available.

doi:10.1371/journal.pcbi.1004365.t001

acquisition and environmental microbial composition. Under our neutral model, microbiome composition is affected by sampling effects. Stochastic changes in microbial abundances may affect the persistence of microbial taxa in the microbiome over one or a few generations (i.e., ecological drift), or over many generations. The latter may occur because host lineages die out; when this happens, changes in microbial abundance across the whole population of hosts are essentially equivalent to changes in allele frequencies (i.e., genetic drift).

The constitution of the microbial community in the environment also plays a considerable role in determining the ultimate fate of microbial taxa within host microbiomes. With a fixed environment, when there is a constant pool of the same microbial taxa from one generation of hosts to the next, microbial taxa never go extinct from the host population as long as hosts obtain some fraction of their microbiome from the environment. This is true, even when that fraction that the environment contributes to each host's microbiome is very small (e.g., 0.1%). In contrast, when the environmental composition of microbes reflects the microbial content of the hosts in previous generations (i.e., PE, the "pooled" environment in our model), microbial diversity of the environment shrinks, as does the diversity of host microbiomes. Therefore, the extent to which parents contribute to the microbiomes of their offspring (either directly or through their contributions to the pooled environment) plays a crucial role in shaping microbiome diversity and constitution. In our simulations, values of  $\alpha$ - and  $\gamma$ - diversities are at their lowest when parental contribution to the microbiome is high. Inevitably, microbial taxa are lost from the population as host lineages are lost.

Thus, under our neutral model, it is possible to recover skewed microbial abundance distributions reminiscent of those obtained with real data, despite a fixed environmental component that remains uniform and constant throughout our simulations. Increasing skewness–essentially, decreasing eveness–is obtained as we increase the degree of parental inheritance. Of course, we don't claim any deep insight here: no one should be surprised that we are able to recover skewed abundance distributions with our models, because there is a large body of literature on the mechanisms–both neutral or otherwise–that may lead to the emergence of skewed abundance distributions (see [46] for an excellent synthesis). Our results reinforce what others [34,47,48] have found, by adding yet another neutral mechanism to account for the emergence of skewed abundance distributions.

Our framework includes sampling effects on an undivided host population, which evolves under a Wright-Fisher process. Consequently, our models have some points of similarity with those that have been developed in population genetics. For instance, Orive et al [49] analyze the evolutionary dynamics of endosymbionts using a discrete-time Moran population genetic model. In their model, endosymbionts are acquired either vertically, passed on from parent to offspring, or horizontally from the environment. This corresponds to our MA x FE model and, in agreement with our results, Orive et al find that increasing the environmental contribution of endosymbionts to host cells results in greater diversity within cells and less diversity between cells.

Our models do not include any mutational process or speciation acting on the microbes, as time moves forward. In reality, of course, microbes acquire mutations in their genomes at a rapid rate, but the measures of diversity we use in our analyses capture differences in taxonomic composition, not genetic diversity. In our models, it is implied that no cladogenetic events have occurred over the course of the simulations.

The models presented here provide an opportunity to construct hypotheses, and make qualitative predictions, about the patterns of diversity we can expect to find in different biological situations. For example, the effects of "pure" pooled versus fixed environments on microbiomes can be found in a comparison of social and solitary bees. Social bees exhibit behaviors that are likely to result in the transmission of microbes from a microbial pool within the colony [50]. In contrast, solitary bees acquire their microbiomes from the environment, through feeding or burrowing. Our model would predict that social bees would have lower  $\alpha$ -diversity and lower taxonomic richness than solitary bees. This is consistent with the results obtained by Martinson et al [51] who surveyed the microbiomes of eusocial bee species *Apis* spp. and *Bombus* spp. and non-social bees (11 species) and wasps (3 species): they found depauperate microbiomes in social bees compared to non-social bees.

Whereas it is reassuring to obtain empirical corroboration for our models, arguably neutral models are most useful when real-world observations run counter to predicted outcomes. Falsification of neutral models provides a justification for augmenting these models to include additional processes that account for the phenomena under study. In this regard, our analysis of the Human Microbiome Project data is instructive. As we have noted above, our simulations should not be used for inference, and we should be cautious about reading too much into the comparisons between empirical and simulated patterns of diversity. Nonetheless, at least for some sites, i.e., the stool, the tongue dorsum, the supragingival plaque, the anterior nares and the buccal mucosa, the low empirical values of  $\alpha$ - and  $\gamma$ - diversities appear to point consistently to a high level of parental inheritance when compared against values obtained in our simulations. There is evidence that the human microbiomes at various sites are seeded at birth by the mother, particularly if this birth is through the vaginal tract [52]. There is also reasonably strong evidence that families share microbes to a greater extent than unrelated individuals in a population [53], and at least in some human populations, mothers share more microbes in common with their offspring than with unrelated children [54]. It is not clear, based on the studies that have been done to date [55], whether the values of direct or indirect parental contribution we obtain when we compare empirical and simulated diversities are significantly higher than would be obtained in real populations, but we expect that the intuition of mosts microbial ecologists is that percentages of direct and pooled parental contributions > 90% are likely to be too high. Putting to one side the caveats about inference, we accept that while this intuition does not constitute evidence against the neutral model, it is likely to engender scepticism about the model's correctness. If it is, in fact, true that direct or indirect parental contributions to the next generation's microbiomes are not as high as our simulations suggest, how do we account for the apparent depression in  $\alpha$ - and  $\gamma$ -diversities, and elevation of  $\beta$ -diversities at these sites? One hypothesis that explains low  $\alpha$ - and  $\gamma$ -diversities, and high  $\beta$ -diversities, and does not require the action of non-neutral processes, is the existence of local host subpopulations. The existence of subpopulations of hosts, with limited immigration and sharing of microbes between subpopulations, is likely to give the appearance of high parental contribution from one generation to the next. Certainly, this is a plausible explanation for patterns of microbiome diversity in the oral cavity (i.e., the buccal mucosa, tongue dorsum and supragingival plaque) and stool samples, because of the likely influence of familial [53] or cultural dietary preferences/practices [56] or lifestyles [57] on these microbiomes. A similarly explanation may account for patterns of microbiome diversity of the anterior nares.

The vaginal posterior fornix presents an interesting contrast to the other body sites because the  $\alpha$ - and  $\gamma$ -diversities suggest high parental contributions (although they cannot distinguish between direct or indirect contributions), whereas  $\beta$ -diversity suggests a low direct parental contribution. This inconsistency may again cause us to reject the neutral model in favor of an alternative explanation, but in this case, subpopulation structure may play a minor role relative to selection for a vaginal microbial community that is common amongst hosts. Such a selective filter is likely a consequence of a complex suite of factors including host immune defences, hormonal cycling, pregnancy, and the presence of apparently beneficial microbial species (e.g., *Lactobacillus* spp.) [58]. This hypothesis explains both the high level of  $\alpha$ - and  $\gamma$ -diversity (i.e., a few abundant species with many rare species), and the low  $\beta$ -diversity. For the human microbiome, neutral models have the potential to help identify additional processes that may account for patterns of diversity. As noted, of the two processes identified above–host subpopulations and selective filters–the former still remains part of an underlying neutral process, and a plausible extension to the neutral framework presented here. Rejection of a simple neutral model therefore allows us to identify incremental additions that may increase explanatory power.

Another example of empirical data that appears to contradict the expectations of our models is the comparison of microbiome diversities in high microbial abundance (HMA) and low microbial abundance (LMA) sponges [59]. HMA sponges have large numbers of associated microbes, in contrast to LMA sponges. Additionally, researchers have shown that microbial diversity in LMA sponges is lower than that of HMA sponges [60,61]. Based on our results, we would predict that there is a greater degree of vertical transmission in LMA sponges, but it turns out that this is not the case: Schmitt et al [62] have found that "vertical transmission, as a mechanism to obtain bacteria, seems to occur mainly in HMA sponges". Giles et al [60] propose two possible reasons to account for the low diversity in LMA sponges. First, there may be selective filters that permit only certain microbial taxa to colonize the sponges; second, the initial colonization event is stochastic, but serves to constrain or exclude successive colonizations. As with the human microbiome data, the sponge example is important because it does not rely a priori on non-neutral processes to account for the low diversity in LMA sponges; instead, selection (or other ecological and/or evolutionary processes) is invoked only after it is shown that vertical transmission in LMA sponges is unlikely, thus indicating that our neutral models are an inadequate explanation for the observed data.

Riffing on the theme that "Essentially, all models are wrong, but some models are useful" [63], Hubbell, writing about models in community ecology, says "Probably no ecologist in the world with even a modicum of field experience would seriously question the existence of niche differences among competing species on the same trophic level" [64]. But, he continues, "[Neutral theory] begins with the simplest possible hypothesis one can think of ... and then adds complexity back into the theory only as absolutely required to obtain satisfactory agreement with the data". We agree with Hubbell: to paraphrase, given what we know about the interplay between hosts, their microbial communities, and the environment, we would hesitate to put money on the table and bet that many microbiomes have evolved under the simplest neutral models that we have constructed here. But we would be equally hesitant betting in favor of the null hypotheses evaluated in statistical tests of significance. The value of these hypotheses resides not in their rightness or wrongness but in their ability to protect against overconfidence in our favorite, more complex model. Whereas it is true that biological processes are frequently complex, Occam's Razor dictates that we construct as simple explanations (or models) as possible. In this way, we remain vigilant against the addition of unnecessary and unjustifiable complexity. Much as we do with statistical hypothesis tests, we accept stronger alternative explanations only when we are sufficiently confident that our neutral hypotheses are unlikely.

This is not to say that neutral models only serve as strawmen; in molecular evolution, for instance, neutral models are frequently effective at explaining molecular variation [65]. And even in cases when the assumption of neutrality is questionable, the use of neutral models of substitution applied in molecular phylogenetics does not appear to jeopardize the accuracy of tree reconstruction [66]. Consequently, without taking account of the evolutionary processes of mutation, speciation, selection or recombination, or the ecological processes that operate in the context of spatial, environmental, and temporal heterogeneity, what we have developed is a framework on which we can begin to evaluate empirical patterns of diversity, and where necessary, add more elaborate ecological and evolutionary scenarios. We believe that even this simple framework, devoid as it is of all the embellishments afforded by evolution and ecology, can

serve a useful purpose: it is a suitable staging ground on which we can construct null models of microbiome diversity in populations of hosts and it allows us to make strong, testable predictions.

#### **Materials and Methods**

Simulated host populations consisted of a fixed number of virtual host individuals (N = 500). Each host was allocated a virtual microbiome with a limited capacity or "slots" of microbes (n = 1000). The environmental pool consisted of 150 microbial taxa. Large number of hosts (N = 2000), microbes (n = 100000) per host and microbial taxa (m = 500) were also simulated with our neutral model, and similar patterns of diversity were observed (S1 Fig).

The microbiomes of the initial generation of host individuals were seeded randomly, with bacteria sampled from a uniform distribution of taxon abundances. We used an initial uniform distribution of taxa because we wanted to ascertain whether the equilibrium distribution of abundances obtained at the conclusion of our simulations would recover patterns seen in natural microbiomes.

For each subsequent generation, the microbiome of each individual host was simulated by populating each of the available "slots" in the individual's microbiome by sampling microbial taxa with replacement (multinomial choice) from either the environment (with probability given (1-x)) or from the microbiome of a "parent" host individual (selected with uniform random probability from the population of the previous generation).

When sampling from a parental/environmental microbial community, the probability that the new host microbiome will acquire a particular microbial taxon is given by the relative abundance of that taxon within the community (see below for details on how environmental microbial taxon abundances were calculated).

The probability, *x*, that a particular "slot" in a new individual host's microbiome was occupied by a microbial taxon sampled from a randomly selected parent was varied across simulations. Two sets of simulations were performed: (1) *x* and *y* varied linearly, between 0 and 1, with increments of 0.1 (see S1–S3 Tables for means and standard deviations of diversities); and (2) with values of  $x, y \in (0.0, 0.5^{[0,1,2...10]})$  (see S4–S6 Tables for diversities). When x = 0.0, a host's microbiome was sampled directly from the environment, i.e., the probability of a microbial taxon being selected was equal to the relative frequency of the microbial taxon in the environment. When  $x = 0.5^0 = 1$ , a host's microbiome was sampled multinomially from the microbiome of its "parent" who was selected with uniform random probability from the previous generation. Similarly, when y = 0.0, there is no contribution from the previous host generation to the environmental microbial resource, whereas when  $y = 0.5^0$ , all microbes are replaced each generation by the pool of microbes resident in the hosts of the previous generation.

Three different models were used to model the relative abundance of taxa in the environment. First, for FE ("fixed environment"), the abundances of the microbial taxa in the environment were fixed to the initial uniform distribution and did not vary over the course of the simulation. Second, under PE ("pooled environment") the abundances of the microbial taxa in the environment were composed of the pooled microbiomes of all the hosts of the previous generation, i.e., by summing over the abundances of the respective microbial taxa in hosts' microbiomes, and renormalizing to relative abundances. Third, under  $ME_y$  ("mixed environment"), the abundances of the microbial taxa in the environment and pooled environment were calculated by combining the fixed environment and pooled environment with y% derived from the pooled environmental component.

A particular simulation regime consisted of a distinct combination of pooled/fixed environmental ratios and environmental factors. A total of ten replicates were run under each simulation regime. The plots of  $\gamma$ -diversity were inspected, and simulations were halted when these stabilized. The number of generations for each simulation varied between  $10^4$  and  $10^6$  generations.

After each generation was simulated, the diversities of microbiomes are measured by scaled Shannon-Wiener index ( $\alpha$ -diversity and  $\gamma$ -diversity) or Bray-Curtis dissimilarity index ( $\beta$ -diversity). The scaled Shannon-Wiener index is calculated as  $-\frac{\sum_{i=1}^{R} p_i ln(p_i)}{\ln R}$ , where

*R* represents the total number of taxa and  $p_i$  represents the relative abundance of *i*th taxon within the community. The calculation of the Bray-Curtis index is given by the formula:  $\frac{2}{(n-1)n} * \sum_{i=2}^{n} \sum_{j=1}^{i-1} \sum_{k=1}^{m} \frac{|p_{ik} - p_{jk}|}{2}$ , where *n* represents the total number of hosts in the population, *m* represents the total number of microbial taxa within the host population, and  $p_{ik}$  and  $p_{jk}$  represents the relative abundance of  $k^{\text{th}}$  taxon within the community of host *i* and host *j*.

Empirical data from the human microbiome were obtained from the website of the NIH Human Microbiome Project (http://www.hmpdacc.org/HMSMCP/). The original community profiling data is a table of relative abundances for each of 690 samples and 718 taxa of bacteria and archaea (from 2 kingdoms to 397 species). As is described on the HMP website, all the samples were collected from 16 body sites from 103 healthy humans and processed with Whole Genome Shotgun sequencing. Specific information of each sample is available on the website (http://www.hmpdacc.org/HMIWGS/all/). We selected data of microbial communities associated with anterior nares, buccal mucosa, supragingival plaque, stool, tongue dorsum and vaginal posterior fornix because of their large numbers of samples, and removed the replicated samples from the same human subject for each body site (see <u>S1 Dataset</u>). The HMP data provides relative abundances for different taxonomic levels. We calculated diversities for all taxonomic levels ranging from species to kingdom, and found that the values of diversities for genus, family and order were similar. Consequently, we chose to use genus-level diversities. The relative abundances of genera for each site and all samples is given in <u>S1 Dataset</u>.

Fitting simulated results into log-normal and Dirichlet-multinomial distributions was performed in R with methods "fitdistr" of package MASS and "dirmult" of package dirmult.

All simulations were carried out using Python scripts and Java programs, available from <a href="https://github.com/qz28/microbiosima.git">https://github.com/qz28/microbiosima.git</a>

Java code: <u>https://github.com/qz28/microbiosima/tree/master/java</u> Python scripts: <u>https://github.com/qz28/microbiosima/tree/master/python</u>

#### **Supporting Information**

S1 Table. α-diversity under different combinations of acquisition and environment models with linear scales for MA(X) and ME(Y). (DOCX)

S2 Table.  $\gamma$ -diversity under different combinations of acquisition and environment models with linear scales for MA(X) and ME(Y). (DOCX)

S3 Table. β-diversity under different combinations of acquisition and environment models with linear scales for MA(X) and ME(Y). (DOCX)

S4 Table. α-diversity under different combinations of acquisition and environment models with power scales for MA(X) and ME(Y). (DOCX)

S5 Table.  $\gamma$ -diversity under different combinations of acquisition and environment models with power scales for MA(X) and ME(Y). (DOCX)

S6 Table. β-diversity under different combinations of acquisition and environment models with power scales for MA(X) and ME(Y). (DOCX)

S1 Dataset. Relative abundance data of empirical human microbial communities associated with anterior nares, buccal mucosa, stool, posterior fornix, tongue dorsum and supragingival plaque from Human Microbiome Project. (XLSX)

S1 Fig. Heatmap of α-, γ-, β-diversity with different combinations of acquisition model and environmental model (Simulated with 2000 hosts, 100000 microbes per host, and 500 microbial taxa until 15000 host generations). The diversity patterns are also represented with heatmaps in a similar way (the proportion of parental inheritance =  $1-0.5^x$  with *x* ranging from 0 to 20; the proportion of pooled environmental component =  $1-0.5^y$  with *y* ranging from 0 to 20), and a similar pattern were still observed regardless of the increasing numbers of hosts, microbes and microbial taxa.

(TIF)

S2 Fig. Heatmap of  $\alpha$ -,  $\gamma$ -,  $\beta$ -diversity reconstructed from Dirichlet-Multinomial distribution. The same log-scales as in Fig 4 is used, and each square represents a re-estimated diversity value from a fitted Dirichlet-Multinomial distribution with ten replicates. (TIF)

#### Acknowledgments

We thank Diana Nemergut, Erin McKenney, and Michael Taylor for constructive comments on the manuscript. Maria Orive and her co-authors kindly shared their manuscript with analytical results that complemented ours. The results of this research were presented on a number of occasions, and this manuscript bears the imprint of comments and suggestions by many people who attended these presentations, including Lawrence David, Yuantong Ding, Raunaq Malhotra, William Parker, Mary Poss, Rytas Vilgalys, and Anne Yoder.

### **Author Contributions**

Conceived and designed the experiments: QZ SW JS AR. Performed the experiments: QZ. Analyzed the data: QZ. Contributed reagents/materials/analysis tools: QZ SW JS AR. Wrote the paper: QZ SW JS AR.

#### References

- 1. Potrikus CJ, Breznak JA (1981) Gut bacteria recycle uric acid nitrogen in termites: A strategy for nutrient conservation. Proc Natl Acad Sci U S A 78: 4601–4605. PMID: <u>16593064</u>
- Zhu L, Wu Q, Dai J, Zhang S, Wei F (2011) Evidence of cellulose metabolism by the giant panda gut microbiome. Proc Natl Acad Sci U S A 108: 17714–17719. doi: <u>10.1073/pnas.1017956108</u> PMID: <u>22006317</u>
- Donia MS, Fricke WF, Partensky F, Cox J, Elshahawi SI, et al. (2011) Complex microbiome underlying secondary and primary metabolism in the tunicate-Prochloron symbiosis. Proc Natl Acad Sci U S A 108: E1423–1432. doi: <u>10.1073/pnas.1111712108</u> PMID: <u>22123943</u>
- 4. Lee FJ, Rusch DB, Stewart FJ, Mattila HR, Newton IL (2014) Saccharide breakdown and fermentation by the honey bee gut microbiome. Environ Microbiol.

- Mendes R, Kruijt M, de Bruijn I, Dekkers E, van der Voort M, et al. (2011) Deciphering the rhizosphere microbiome for disease-suppressive bacteria. Science 332: 1097–1100. doi: <u>10.1126/science</u>. 1203980 PMID: 21551032
- Sarmiento-Ramirez JM, van der Voort M, Raaijmakers JM, Dieguez-Uribeondo J (2014) Unravelling the microbiome of eggs of the endangered sea turtle Eretmochelys imbricata identifies bacteria with activity against the emerging pathogen Fusarium falciforme. PLoS One 9: e95206. doi: <u>10.1371/</u> journal.pone.0095206 PMID: 24743166
- 7. Rolli E, Marasco R, Vigani G, Ettoumi B, Mapelli F, et al. (2014) Improved plant resistance to drought is promoted by the root-associated microbiome as a water stress-dependent trait. Environ Microbiol.
- Venu I, Durisko Z, Xu J, Dukas R (2014) Social attraction mediated by fruit flies' microbiome. J Exp Biol 217: 1346–1352. doi: <u>10.1242/jeb.099648</u> PMID: <u>24744425</u>
- Ohland CL, Kish L, Bell H, Thiesen A, Hotte N, et al. (2013) Effects of Lactobacillus helveticus on murine behavior are dependent on diet and genotype and correlate with alterations in the gut microbiome. Psychoneuroendocrinology 38: 1738–1747. doi: <u>10.1016/j.psyneuen.2013.02.008</u> PMID: <u>23566632</u>
- Ezenwa VO, Gerardo NM, Inouye DW, Medina M, Xavier JB (2012) Microbiology. Animal behavior and the microbiome. Science 338: 198–199. doi: 10.1126/science.1227412 PMID: 23066064
- Cryan JF, O'Mahony SM (2011) The microbiome-gut-brain axis: from bowel to behavior. Neurogastroenterol Motil 23: 187–192. doi: 10.1111/j.1365-2982.2010.01664.x PMID: 21303428
- Allegretti JR, Hamilton MJ (2014) Restoring the gut microbiome for the treatment of inflammatory bowel diseases. World J Gastroenterol 20: 3468–3474. doi: <u>10.3748/wjg.v20.i13.3468</u> PMID: <u>24707129</u>
- Mondot S, de Wouters T, Dore J, Lepage P (2013) The human gut microbiome and its dysfunctions. Dig Dis 31: 278–285. doi: 10.1159/000354678 PMID: 24246975
- Morgan XC, Tickle TL, Sokol H, Gevers D, Devaney KL, et al. (2012) Dysfunction of the intestinal microbiome in inflammatory bowel disease and treatment. Genome Biol 13: R79. doi: <u>10.1186/gb-2012-13-</u> <u>9-r79</u> PMID: <u>23013615</u>
- Foster JA, McVey Neufeld KA (2013) Gut-brain axis: how the microbiome influences anxiety and depression. Trends Neurosci 36: 305–312. doi: 10.1016/j.tins.2013.01.005 PMID: 23384445
- 16. Ley RE (2010) Obesity and the human microbiome. Curr Opin Gastroenterol 26: 5–11. doi: 10.1097/ MOG.0b013e328333d751 PMID: 19901833
- Turnbaugh PJ, Ley RE, Mahowald MA, Magrini V, Mardis ER, et al. (2006) An obesity-associated gut microbiome with increased capacity for energy harvest. Nature 444: 1027–1031. PMID: <u>17183312</u>
- Mulle JG, Sharp WG, Cubells JF (2013) The gut microbiome: a new frontier in autism research. Curr Psychiatry Rep 15: 337. doi: <u>10.1007/s11920-012-0337-0</u> PMID: <u>23307560</u>
- 19. Macfabe DF (2012) Short-chain fatty acid fermentation products of the gut microbiome: implications in autism spectrum disorders. Microb Ecol Health Dis 23.
- 20. Inoue Y, Shimojo N (2014) Microbiome/microbiota and allergies. Semin Immunopathol.
- Dickson RP, Martinez FJ, Huffnagle GB (2014) The role of the microbiome in exacerbations of chronic lung diseases. Lancet 384: 691–702. doi: <u>10.1016/S0140-6736(14)61136-3</u> PMID: <u>25152271</u>
- Segal LN, Alekseyenko AV, Clemente JC, Kulkarni R, Wu B, et al. (2013) Enrichment of lung microbiome with supraglottic taxa is associated with increased pulmonary inflammation. Microbiome 1: 19. doi: <u>10.1186/2049-2618-1-19</u> PMID: <u>24450871</u>
- Han MK, Huang YJ, Lipuma JJ, Boushey HA, Boucher RC, et al. (2012) Significance of the microbiome in obstructive lung disease. Thorax 67: 456–463. doi: <u>10.1136/thoraxjnl-2011-201183</u> PMID: <u>22318161</u>
- Proctor LM (2011) The Human Microbiome Project in 2011 and beyond. Cell Host Microbe 10: 287– 291. doi: <u>10.1016/j.chom.2011.10.001</u> PMID: <u>22018227</u>
- Qin J, Li R, Raes J, Arumugam M, Burgdorf KS, et al. (2010) A human gut microbial gene catalogue established by metagenomic sequencing. Nature 464: 59–65. doi: <u>10.1038/nature08821</u> PMID: <u>20203603</u>
- 26. Gajer P, Brotman RM, Bai G, Sakamoto J, Schütte UM, et al. (2012) Temporal dynamics of the human vaginal microbiota. Science translational medicine 4: 132ra152–132ra152.
- Koenig JE, Spor A, Scalfone N, Fricker AD, Stombaugh J, et al. (2011) Succession of microbial consortia in the developing infant gut microbiome. Proc Natl Acad Sci U S A 108 Suppl 1: 4578–4585. doi: <u>10.</u> <u>1073/pnas.1000081107</u> PMID: <u>20668239</u>
- Grice EA, Kong HH, Conlan S, Deming CB, Davis J, et al. (2009) Topographical and temporal diversity of the human skin microbiome. Science 324: 1190–1192. doi: <u>10.1126/science.1171700</u> PMID: <u>19478181</u>

- Claus SP, Tsang TM, Wang Y, Cloarec O, Skordi E, et al. (2008) Systemic multicompartmental effects of the gut microbiome on mouse metabolic phenotypes. Mol Syst Biol 4: 219. doi: <u>10.1038/msb.2008.</u> 56 PMID: 18854818
- Costello EK, Stagaman K, Dethlefsen L, Bohannan BJ, Relman DA (2012) The application of ecological theory toward an understanding of the human microbiome. Science 336: 1255–1262. doi: <u>10.1126/</u> science.1224203 PMID: 22674335
- Yeoman CJ, Chia N, Yildirim S, Miller MEB, Kent A, et al. (2011) Towards an evolutionary model of animal-associated microbiomes. Entropy 13: 570–594.
- Leibold MA, Holyoak M, Mouquet N, Amarasekare P, Chase JM, et al. (2004) The metacommunity concept: a framework for multi-scale community ecology. Ecology Letters 7: 601–613.
- Cadotte MW (2006) Metacommunity influences on community richness at multiple spatial scales: a microcosm experiment. Ecology 87: 1008–1016. PMID: <u>16676544</u>
- **34.** Hubbell SP (2001) The unified neutral theory of biodiversity and biogeography (MPB-32): Princeton University Press.
- Hellweger FL, van Sebille E, Fredrick ND (2014) Biogeographic patterns in ocean microbes emerge in a neutral agent-based model. Science 345: 1346–1349. doi: <u>10.1126/science.1254421</u> PMID: 25214628
- Kingman JFC (1982) On the Genealogy of Large Populations. Journal of Applied Probability 19: 27– 43.
- 37. King JL, Jukes TH (1969) Non-Darwinian evolution. Science 164: 788–798. PMID: 5767777
- Alonso D, Etienne RS, McKane AJ (2006) The merits of neutral theory. Trends Ecol Evol 21: 451–457. PMID: <u>16766082</u>
- 39. Wright S (1931) Evolution in Mendelian populations. Genetics 16: 97–159. PMID: 17246615
- 40. Fisher RA (1922) On the dominance ratio. Proceedings of the Royal Society, Edinburgh 42: 321–341.
- La Rosa PS, Brooks JP, Deych E, Boone EL, Edwards DJ, et al. (2012) Hypothesis testing and power calculations for taxonomic-based human microbiome data. PLoS One 7: e52078. doi: <u>10.1371/journal.pone.0052078</u> PMID: <u>23284876</u>
- Chesson P, Huntly N (1997) The roles of harsh and fluctuating conditions in the dynamics of ecological communities. Am Nat 150: 519–553. doi: 10.1086/286080 PMID: 18811299
- Peterson J, Garges S, Giovanni M, McInnes P, Wang L, et al. (2009) The NIH Human Microbiome Project. Genome Res 19: 2317–2323. doi: 10.1101/gr.096651.109 PMID: 19819907
- 44. Moeller AH, Li Y, Mpoudi Ngole E, Ahuka-Mundeke S, Lonsdorf EV, et al. (2014) Rapid changes in the gut microbiome during human evolution. Proc Natl Acad Sci U S A 111: 16431–16435. doi: <u>10.1073/pnas.1419136111</u> PMID: <u>25368157</u>
- Stumpf RM, Wilson BA, Rivera A, Yildirim S, Yeoman CJ, et al. (2013) The primate vaginal microbiome: comparative context and implications for human health and disease. Am J Phys Anthropol 152 Suppl 57: 119–134. doi: 10.1002/ajpa.22395 PMID: 24166771
- McGill BJ, Etienne RS, Gray JS, Alonso D, Anderson MJ, et al. (2007) Species abundance distributions: moving beyond single prediction theories to integration within an ecological framework. Ecol Lett 10: 995–1015. PMID: <u>17845298</u>
- Bell G (2000) The Distribution of Abundance in Neutral Communities. Am Nat 155: 606–617. PMID: 10777433
- Caswell H (1976) Community structure: a neutral model analysis. Ecological Monographs 46: 327– 354.
- 49. Orive ME, Monnahan P, Rong S, Wakeley J (In preparation) A new model of host-endosymbiont population dynamics.
- Powell JE, Martinson VG, Urban-Mead K, Moran NA (2014) Routes of acquisition of the gut microbiota of Apis mellifera. Appl Environ Microbiol.
- Martinson VG, Danforth BN, Minckley RL, Rueppell O, Tingek S, et al. (2011) A simple and distinctive microbiota associated with honey bees and bumble bees. Mol Ecol 20: 619–628. doi: <u>10.1111/j.1365-294X.2010.04959.x</u> PMID: <u>21175905</u>
- 52. Dominguez-Bello MG, Costello EK, Contreras M, Magris M, Hidalgo G, et al. (2010) Delivery mode shapes the acquisition and structure of the initial microbiota across multiple body habitats in newborns. Proceedings of the National Academy of Sciences 107: 11971–11975.
- Song SJ, Lauber C, Costello EK, Lozupone CA, Humphrey G, et al. (2013) Cohabiting family members share microbiota with one another and with their dogs. Elife 2: e00458. doi: <u>10.7554/eLife.00458</u> PMID: <u>23599893</u>

- Yatsunenko T, Rey FE, Manary MJ, Trehan I, Dominguez-Bello MG, et al. (2012) Human gut microbiome viewed across age and geography. Nature 486: 222–227. doi: <u>10.1038/nature11053</u> PMID: <u>22699611</u>
- Spor A, Koren O, Ley R (2011) Unravelling the effects of the environment and host genotype on the gut microbiome. Nature Reviews Microbiology 9: 279–290. doi: 10.1038/nrmicro2540 PMID: 21407244
- Adler CJ, Dobney K, Weyrich LS, Kaidonis J, Walker AW, et al. (2013) Sequencing ancient calcified dental plaque shows changes in oral microbiota with dietary shifts of the Neolithic and Industrial revolutions. Nature genetics 45: 450–455. doi: <u>10.1038/ng.2536</u> PMID: <u>23416520</u>
- 57. Sampaio-Maia B, Monteiro-Silva F (2014) Acquisition and maturation of oral microbiome throughout childhood: An update. Dent Res J (Isfahan) 11: 291–301.
- **58.** Witkin S (2014) The vaginal microbiome, vaginal anti-microbial defence mechanisms and the clinical challenge of reducing infection-related preterm birth. BJOG.
- Hentschel U, Fieseler L, Wehrl M, Gernert C, Steinert M, et al. (2003) Microbial diversity of marine sponges. In: Mueller W, editor. Marine Molecular Biotechnology. Berlin: Springer. pp. 59–88.
- 60. Giles EC, Kamke J, Moitinho-Silva L, Taylor MW, Hentschel U, et al. (2013) Bacterial community profiles in low microbial abundance sponges. FEMS Microbiol Ecol 83: 232–241. doi: <u>10.1111/j.1574-6941.2012.01467.x</u> PMID: <u>22882238</u>
- Kamke J, Taylor MW, Schmitt S (2010) Activity profiles for marine sponge-associated bacteria obtained by 16S rRNA vs 16S rRNA gene comparisons. ISME J 4: 498–508. doi: <u>10.1038/ismej.2009.143</u> PMID: <u>20054355</u>
- 62. Schmitt S, Wehrl M, Lindquist N, Weisz JB, Hentschel U (2007) Morphological and molecular analyses of microorganisms in Caribbean reef adult sponges and in corresponding reproductive material. In: Custódio MR, Lôbo-Hajdu G, Hajdu E, Muricy G, editors. Porifera Research: Biodiversity, Innovation and Sustainability. Rio de Janeiro: Museu Nacional. pp. 561–568.
- 63. Box GEP, Draper NR (1987) Empirical Model Building and Response Surfaces. New York: John Wiley and Sons.
- **64.** Hubbell SP (2005) Neutral theory in community ecology and the hypothesis of functional equivalence. Functional Ecology 19: 166–172.
- Nei M, Suzuki Y, Nozawa M (2010) The neutral theory of molecular evolution in the genomic era. Annu Rev Genomics Hum Genet 11: 265–289. doi: <u>10.1146/annurev-genom-082908-150129</u> PMID: 20565254
- 66. Roje DM (2014) Evaluating the effects of non-neutral molecular markers on phylogeny inference. PLoS One 9: e87428. doi: 10.1371/journal.pone.0087428 PMID: 24558367